**Author's response to reviews**

**Title:** Malignant fibrous hisriocytoma or the face: report of a case.

**Authors:**

Laszlo Seper (seper@uni-muenster.de)
Richard Schwab (schwab@kkk.org.hu)
Sirichai Kiattavorncharoen2 (kiattav@hotmail.com)
Andre Buchter (a.buechter@citykom.net)
Agnes Bankfalvi (piffko@uni-muenster.de)
Ulrich Joos (iapos@uni-muenster.de)
Jozsef Piffko (piffko@uni-muenster.de)
Birgit Kruse-Loesler (losler@uni-munster.de)

**Version:** 2  **Date:** 7 March 2007

**Author's response to reviews:**

1) The organisation of the Abstract was change as follows: Background, Case report and Discussion.
2) The background section in the Abstract was completed with a statement about the aim of the paper.
3) The organisation of the paper was change as follows: Background, Case report and Discussion
4) The background section of the paper was completed with a statement about the aim of the case report.